Author's response to reviews

Title: Khat use as risk factor for psychotic disorders: A cross-sectional and case-control study in Somalia

Authors:

Michael Odenwald (michael.odenwald@uni-konstanz.de)
Frank Neuner (frank.neuner@uni-konstanz.de)
Margarete Schauer (margarete.schauer@uni-konstanz.de)
Thomas R Elbert (thomas.elbert@uni-konstanz.de)
Claudia Catani (claudia.catani@uni-konstanz.de)
Birke Lingenfelder (birke.lingenfelder@uni-konstanz.de)
Harald Hinkel (haraldhinkel@hotmail.com)
Heinz Hafner (hhaefner@as200.zi-mannheim.de)
Brigitte Rockstroh (brigitte.rockstroh@uni-konstanz.de)

Version: 3 Date: 10 January 2005

Author's response to reviews:

Authors' qualifications removed from title page.
Authors' affiliations corrected on cover page, according to instruction.
Figure and table placement text removed.
Competing interest declaration changed in wording to ‘The authors declare that they have no competing interests’.
Acknowledgements placed after authors' contributions.
Shading in tables removed, highlighting in tables by making them bold.
Changes in text, as follows:
Semicolons and spelling errors corrected.
According to the reviewer's suggestion reference to UNODC added to introduction: ‘...This is reflected in the most recent issue of the annual World Drug Report: in 2001 all countries, where khat is considered to be the main problem drug (Eritrea, Ethiopia, Djibouti, Yemen, Somalia), report an increase in its use, in 2002 an increase was found in four of them (UNODC, 2004).’